Author's response to reviews

Title: Clinical and prognostic significance of preoperative plasma hyperfibrinogenemia in gallbladder cancer following surgical resection

Authors:

Yi-jun Shu (shuyijun19881125@163.com)
Hao Weng (535836591@qq.com)
Run-Fa Bao (baorunfa163@163.com)
Xiang-Song Wu (wuxiangsong_1983@sina.com)
Qian Ding (dqiana@gmail.com)
Yang Cao (cycaoyang@163.com)
Xu-An Wang (xu_anwang@126.com)
Fei Zhang (fei_zhang126.com)
Shan-Shan Xiang (843178159@qq.com)
Huai-Feng Li (lhf_ok@163.com)
Mao-Lan Li (limaolan6@163.com)
Jia-Sheng Mu (good_surgeon@hotmail.com)
Wen-Guang Wu (wwengguang08@126.com)
Ying-Bin Liu (liuybphd@126.com)

Version: 5 Date: 28 July 2014

Author's response to reviews:

Dear editor:

Thank you for considering our work Manuscript MS: 1254132792129466 entitled "Clinical and prognostic significance of preoperative plasma hyperfibrinogenemia in gallbladder cancer following surgical resection". Your suggestions are helpful and make our research more comprehensive. According to your recommend, We have asked a professional language editing service from Edanz. They have copyedited our manuscript by a native-English speaker with scientific expertise. The revised manuscript has been uploaded. All the changes in the manuscript have been highlighted. Meanwhile, language editing certification from Edanz and trackedcopy are submitted to you in the mail attachment. We tried our best to make this revision as soon as possible and we hope it will meet the editorial requirements and to be published in BMC Cancer.

Sincerely yours,

Yingbin Liu, Ph.D & M.D
Department of General Surgery, Xinhua Hospital, Affiliated to School of Medicine, Shanghai Jiao Tong University
1665 Kongjiang Road,
Shanghai 200092, P.R.China